Author's response to reviews

Title: PTPRC (CD45) is not associated with multiple sclerosis in a large cohort of German patients

Authors:

Bianca B Miterski (bianca.miterski@ruhr-uni-bochum.de)
Eckhart E Sindern (eckhart.sindern@ruhr-uni-bochum.de)
Michael M Haupts (michael.haupts@ruhr-uni-bochum.de)
Sebastian S Schimrigk (sebastian.k.schimrigk@ruhr-uni-bochum.de)
Joerg JT Epplen (joerg.t.epplen@ruhr-uni-bochum.de)

Version: 2 Date: 13 Feb 2002

Results and conclusions: ..., even when regarding the subgroups of patients with different clinical courses of MS (RRMS = relapsing remitting course of MS; PPMS = primary progressive MS; SPMS = secondary progressive MS). Additional data are found in Table 1.